Author’s response to reviews

Title: Auto-titrating Continuous Positive Airway Pressure Treatment for Obstructive Sleep Apnoea after Acute Quadriplegia (COSAQ): study protocol for a randomized controlled trial

Authors:

David J Berlowitz (david.berlowitz@austin.org.au)
Najib Ayas (NAyas@providencehealth.bc.ca)
Maree Barnes (maree.barnes@austin.org.au)
Douglas j Brown (Douglas.J.Brown@austin.org.au)
Peter Cistulli (peter.cistulli@sydney.edu.au)
Tim Geraghty (Timothy_Geraghty@health.qld.gov.au)
Alison Graham (Alison.Graham@buckshealthcare.nhs.uk)
Bonne B Lee (Bonne.Lee@SESIAHS.HEALTH.NSW.GOV.AU)
Meg Morris (M.Morris@latrobe.edu.au)
Fergal O'Donoghue (Fergal.O'Donoghue@austin.org.au)
Peter D Rochford (peter.rochford@austin.org.au)
Jack Ross (jack.ross@austin.org.au)
Raj Singhal (Raj.Singhal@cchb.govt.nz)
Jo Spong (jo.spong@austin.org.au)
Brooke Wadsworth (brook_wadsworth@health.qld.gov.au)
Robert J Pierce (xx@xx.com)

Version: 5 Date: 21 May 2013

Author’s response to reviews:

See attached